Author's response to reviews

Title: Clinically relevant safety issues associated with St. John’s wort product labels

Authors:

Kevin A Clauson (clauson@nsu.nova.edu)
Marile L Santamarina (marile_santamarina@pba.edu)
Jennifer C Rutledge (jrutledg@nsu.nova.edu)

Version: 4 Date: 1 July 2008

Author's response to reviews: see over
Response of the authors to comments by the reviewers and editor(s)

Title: Clinically relevant safety issues associated with St. John’s wort product labels (Manuscript ID 2050231122188761)

Authors:

KAC: clauson@nsu.nova.edu
MLS: Marile_Santamarina@pba.edu
JCR: jrutledg@nsu.nova.edu
Iraxte Puebla  
Senior Assistant Editor  
BMC Complementary and Alternative Medicine  

1JUL08  

Dr. Puebla,  

We are providing a revised copy of the manuscript, “Clinically relevant safety issues associated with St. John’s wort product labels” (Manuscript ID 2050231122188761). Based on the changes we made, it now fully conforms to the editor’s requests.

Specifically, we have reformatted the references in the body of the manuscript. We have also removed the Figures along with the textual references to them. However, we still maintain that there is a clear utility for the inclusion of (at least) Figure 1 or an embedded thumbnail image of each supplement in Table 1.

Regardless, we appreciate your assistance during the process and are glad to have received your notification of acceptance of our article for publication in BMC Complementary and Alternative Medicine.

Sincerely,

Kevin A. Clauson, Pharm.D.